Author's response to reviews

Title: The asthma candidate gene NPSR1 mediates isoform specific downstream signalling

Authors:

Christina Orsmark Pietras (christina.orsmark-pietras@ki.se)
Johanna Vendelin (johanna.vendelin@gmail.com)
Francesca Anedda (fane2004@gmail.com)
Sara Bruce (sara.bruce@ki.se)
Mikael Adner (mikael.adner@ki.se)
Lilli Sundman (lilli.sundman@helsinki.fi)
Ville Pulkkinen (ville.pulkkinen@helsinki.fi)
Harri Alenius (harri.alenius@ttl.fi)
Mauro D'Amato (Mauro.Damato@ki.se)
Cilla Söderhäll (Cilla.Soderhall@ki.se)
Juha Kere (Juha.Kere@ki.se)

Version: 5 Date: 10 May 2011

Author's response to reviews: see over
Dear Prof Kere,

Thank you for submitting your revised manuscript to us. I sincerely apologise for any delay, however before we can proceed further we must ask for more information regarding the deposition of your data.

In your cover letter you provide us with a log-on and password to access your deposited data and assured us that the data will be made publicly available after acceptance of your manuscript.

We are grateful for this information, however please note that it is a requirement of our editorial policies at BioMed Central that your data must be made publicly available BEFORE we can accept your manuscript.

This point must be addressed before we can proceed further.

Please make the data publicly available and revise your manuscript to ensure that sufficient detail is included to allow readers to find and access your data.

Once this issue has been resolved, we anticipate rapid acceptance of your manuscript.

We understand your wish for your manuscript to be published as soon as possible. Therefore I would advise you to contact me at tim.shipley@biomedcentral.com once you have uploaded your revised manuscript onto our system.

This will ensure that your revised manuscript is assessed as soon as possible.

If you have any questions, please contact me using the same email address.

We would be grateful if you could address the comments in a revised manuscript and provide a cover letter giving a point-by-point response to the concerns.

Please also highlight (with 'tracked changes'/coloured/underlines/highlighted text) all changes made when revising the manuscript to make it easier for the Editors to give you a prompt decision on your manuscript.

Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals ). It is important that your files are correctly formatted.

We look forward to receiving your revised manuscript by 9 June 2011. If you imagine that it will take longer to prepare please give us some estimate of when we can expect it.

You should upload your cover letter and revised manuscript through http://www.biomedcentral.com/manuscript/login/man.asp?txt_nav=man&txt_man_id=1443247055467995. You will find more detailed instructions at the base of this email.
Dear Editor,

We do apologise for the misunderstanding regarding the release of the micro array data. The data, available at http://www.ebi.ac.uk/arrayexpress/ with the The ArrayExpress accession: E-MEXP-2782 will be publically released as of tomorrow 11 May2011. Sufficient information how to access the data is now available to the readers in the manuscript (changes marked in yellow).

We also greatly appreciate your understanding of our wish for a rapid acceptance of the manuscript.

Sincere regards,

Juha Kere
Corresponding author